Author's response to reviews

Title: The effect of tracheotomy on drug consumption in patients with acute aneurysmal subarachnoid hemorrhage.

Authors:

Leiv A Rosseland (l.a.rosseland@medisin.uio.no)
Jon Narum (jnarum@ous-hf.no)
Audun Stubhaug (audun.stubhaug@medisin.uio.no)
Ulf Kongsgaard (ulf.kongsgaard@medisin.uio.no)
Wilhelm Sorteberg (wilhelm.sorteberg@ous-hf.no)
Angelika Sorteberg (angelika.sorteberg@ous-hf.no)

Version: 2
Date: 26 September 2014

Author's response to reviews:

Dear Editor,

Thank you for reviewing the manuscript.

The following changes have been made:

We have included line numbers and page numbers.

Abbreviation list, authors competing interests, and authors contributions have been declared.

Sincerely, Leiv A Rosseland